Author's response to reviews

Title: Exploring the oral microbiota of children at various developmental stages of their dentition in the relation to their oral health

Authors:

Wim Crielaard (W.Crielaard@acta.nl)
Egija Zaura (E.Zaura@acta.nl)
Annemarie A Schuller (Annemarie.Schuller@tno.nl)
Sue M Huse (shuse@mbl.edu)
Roy C Montijn (Roy.Montijn@tno.nl)
Bart JF Keijser (Bart.Keijser@tno.nl)

Version: 3 Date: 28 February 2011

Author's response to reviews:

This version contains the formatting changes as requested